Author’s response to reviews

Title: Biomarker Screening for Antenatal Depression in Women Who Underwent Caesarean Section: A Matched Observational Study with Plasma Lipidomics

Authors:

Zhuoxi Wu (WUSISSI527@163.com)
Peng Zhao (66949366@qq.com)
Zhonghong Long (FANNLZH@163.com)
Jie Li (824679901@qq.com)
Guiying Yang (472151685@qq.com)
Qingling Zhang (2805604446@qq.com)
Guangyou Duan (dgy1986anesthesia@126.com)
Hong Li (lh78553@163.com)

Version: 4 Date: 11 Aug 2019

Author’s response to reviews:

12th Aug. 2019

BMC Psychiatry

Dear editor

I along with my co-authors would like to re-submit the manuscript entitled “Biomarkers Screening for Antenatal Depression in Caesarean Women: A Matched Observational Study with Plasma Lipidomics” to BMC Psychiatry. The previous manuscript ID is BPSY-D-18-00982R1.

Thank you again for the opportunity to revise the manuscript. We had improved the language of manuscript through a professional English language editing company. The manuscript has been carefully rechecked, and appropriate changes have been made.
Editor Comments:

1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication.
Respond: We had improved the language of manuscript through a professional English language editing company. It took some time to carefully recheck the details of the revision. We apologize for the delay and thank you for the opportunity to revise the manuscript again.

2. Please upload your revised manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. Please also ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.
Respond: We have submitted a revision manuscript in a clean version as requested by the editor.

We hope that the revised manuscript is now suitable for publication in your journal.

I look forward to your reply.

Sincerely,

Hong Li, Director, Department of Anesthesiology, Xinqiao Hospital, Third Military Medical University, Chongqing, 400037, China.
Telephone: (+86)13608380123; Email: lh78553@163.com.